

June 30, 2023

THIS TRANSMITTAL CONTAINS IMPORTANT INFORMATION THAT IS OF INTEREST TO THE BENEFICIAL OWNERS OF THE NOTES. IF APPLICABLE, ALL DEPOSITORIES, CUSTODIANS AND OTHER INTERMEDIARIES RECEIVING THIS NOTICE ARE REOUESTED TO EXPEDITE RE-TRANSMITTAL TO SUCH BENEFICIAL OWNERS IN A TIMELY MANNER.

## DRYDEN 87 CLO, LTD. **DRYDEN 87 CLO, LLC**

## NOTICE OF EXECUTED SUPPLEMENTAL INDENTURE

Holders of Notes issued by Dryden 87 CLO, Ltd. and Dryden 87 CLO, LLC and the To: Addressees listed in Exhibit 2 attached hereto.

> (Classes and CUSIP numbers1 are listed on Exhibit 1 attached hereto and Addressees are listed on Exhibit 2 attached hereto)

Ladies and Gentlemen:

Reference is made to the Indenture, dated as of July 7, 2021 (as amended, supplemented or otherwise modified from time to time and as in effect on the date hereof, the "Indenture") among Dryden 87 CLO, Ltd., as issuer (the "Issuer"), Dryden 87 CLO, LLC, as co-issuer (the "Co-Issuer" and, together with the Issuer, the "Co-Issuers"), and Deutsche Bank Trust Company Americas, as trustee (in such capacity, the "Trustee"). Capitalized terms used but not defined herein have the meanings ascribed to them in the Indenture.

The Trustee hereby provides notice to all Holders and the addressees on Exhibit 2 hereto that the Issuers and the Trustee entered into a First Supplemental Indenture. A copy of the executed First Supplemental Indenture is attached hereto as Exhibit 3.

Please contact Rick Kohlmeyer or Sun-Hee Chang at Deutsche Bank Trust Company Americas for any questions regarding this notice. Rick Kohlmeyer can be contacted at 714.247.6339 or rick.kohlmeyer@db.com and Sun-Hee Chang can be contacted at 714.247.6146 or sun-hee.chang@db.com.

> DEUTSCHE BANK TRUST COMPANY AMERICAS. as Trustee

<sup>&</sup>lt;sup>1</sup> CUSIP numbers are included solely for the convenience of the Holders. The Trustee is not responsible for the selection or use of the CUSIP numbers, or the accuracy of any such numbers printed on the Notes or indicated in this notice.

## Exhibit 1

Class	CUSIP
Class A-1 Notes 144A	26246GAA1
Class A-1 Notes REG S	G2857RAA1
Class A-2 Notes 144A	26246GAC7
Class A-2 Notes REG S	G2857RAB9
Class B Notes 144A	26246GAE3
Class B Notes REG S	G2857RAC7
Class C Notes 144A	26246GAG8
Class C Notes REG S	G2857RAD5
Class D Notes 144A	26246GAJ2
Class D Notes REG S	G2857RAE3
Class E Notes 144A	26246HAA9
Class E Notes REG S	G2857VAA2
Subordinated Notes 144A	26246HAC5
Subordinated Notes REG S	G2857VAB0

### Exhibit 2

Dryden 87 CLO, Ltd. c/o MaplesFS Limited P.O. Box 1093 Boundary Hall, Cricket Square Grand Cayman, KY1-1102 Cayman Islands cayman@maples.com

Dryden 87 CLO, LLC c/o Puglisi & Associates 850 Library Avenue, Suite 204 Newark, Delaware 19711 dpuglisi@puglisiassoc.com

PGIM, Inc. 655 Broad Street Newark, New Jersey 07102-4410 Attention: Managing Director, CLO Unit bent.hoyer@pgim.com

Cayman Islands Stock Exchange PO Box 2408 Grand Cayman KY1-1105 Cayman Islands listing@csx.ky

S&P Global Ratings
55 Water Street, 41st Floor
New York, New York 10041-0003
Attention: Asset Backed-CBO/CLO Surveillance
CDO\_Surveillance@spglobal.com

For posting on the 17g-5 website (NRSRO Website) Email: SEC.17g-5@db.com; dryden\_usa@list.db.com

## Exhibit 3

[Executed First Supplemental Indenture]

## FIRST SUPPLEMENTAL INDENTURE

among

DRYDEN 87 CLO, LTD. as Issuer

DRYDEN 87 CLO, LLC as Co-Issuer

and

DEUTSCHE. BANK TRUST COMPANY AMERICAS as Trustee

First Supplemental Indenture, dated as of June 30, 2023 (supplement to the Indenture dated as of July 7, 2021)

THIS FIRST SUPPLEMENTAL INDENTURE (this "Supplemental Indenture"), dated as of June 30, 2023, among Dryden 87 CLO, Ltd., an exempted company incorporated with limited liability under the laws of the Cayman Islands (the "Issuer"), DRYDEN 87 CLO, LLC, a limited liability company formed under the laws of the State of Delaware (the "Co-Issuer" and, together with the Issuer, the "Co-Issuers"), and Deutsche Bank Trust Company Americas, as trustee (herein, together with its permitted successors and assigns in the trusts hereunder, the "Trustee"), hereby amends the Indenture dated as of July 7, 2021 (as supplemented or amended from time to time, the "Indenture"), among the Issuer, the Co-Issuer and the Trustee. Capitalized terms used in this Supplemental Indenture that are not otherwise defined herein have the meanings assigned thereto in the Indenture.

### WITNESSETH

WHEREAS, pursuant to the applicable provisions of the Indenture, if the Collateral Manager determines that a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of the determination of the then-current Benchmark on any date, then upon written notice by the Collateral Manager to the Issuer, the Calculation Agent, the Collateral Administrator and the Trustee, the Alternative Reference Rate will replace the then-current Benchmark for all purposes relating to the Floating Rate Notes in respect of such determination on such date and all determinations on subsequent dates;

WHEREAS, the Collateral Manager has provided such notice to the effect that a Benchmark Transition Event and its related Benchmark Replacement Date will occur on June 30, 2023, and following the Benchmark Transition Event and Benchmark Replacement Date occurring on June 30, 2023, the Alternative Reference Rate will replace LIBOR as the Benchmark;

WHEREAS, a supplemental indenture is not required in order to adopt an Alternative Reference Rate;

WHEREAS, the Benchmark Replacement is the Alternative Reference Rate if (i) the Collateral Manager determines such rate can be determined as of the Benchmark Replacement Date and (ii) such rate is being used by at least 50% of the Aggregate Principal Amount of the Floating Rate Collateral Debt Obligations included in the Trust Estate that pay interest quarterly; however, the latter condition is not satisfied as of the effective date of this Supplemental Indenture;

WHEREAS, such Benchmark Replacement would be the sum of Term SOFR for the applicable index and the applicable Benchmark Replacement Adjustment;

WHEREAS, the Relevant Governmental Body has recommended that the applicable spread adjustment for three-month Term SOFR is 0.26161%;

WHEREAS, if the Benchmark Replacement cannot be determined by the Collateral Manager as of the Benchmark Replacement Date, then the Alternative Reference Rate is the first alternative in the following order that the Collateral Manager determines can be determined as of the Benchmark Replacement Date: (1) the rate proposed by the Collateral Manager and consented to by a Majority of the Controlling Class and a Majority of the Subordinated Notes; and (2) the

Fallback Rate; provided that the Alternative Reference Rate for the Floating Rate Notes will be no less than zero;

WHEREAS, the Collateral Manager has proposed the sum of (a) Term SOFR (for an index maturity of three months) <u>plus</u> (b) 0.26161% to be both (1) the rate for which consent was requested from a Majority of the Controlling Class and a Majority of the Subordinated Notes and (2) the applicable Fallback Rate determined in its commercially reasonable discretion if such consents were not received;

WHEREAS, such consents were requested but not timely received, and therefore the Alternative Reference Rate will be the above rate as determined by the Collateral Manager in its commercially reasonable discretion to be the Fallback Rate;

WHEREAS, pursuant to Section 8.1(a)(xxxvii) of the Indenture, without the consent of the Noteholders, but with the prior written consent of the Collateral Manager, the Co-Issuers and the Trustee may enter into one or more supplemental indentures to implement any Benchmark Replacement Conforming Changes;

WHEREAS, pursuant to Section 8.1 of the Indenture, a proposed supplemental indenture to implement Benchmark Replacement Conforming Changes may be effected without prior notice to the Noteholders, any Hedge Counterparty or each Rating Agency provided that as soon as practicable following execution of such supplemental indenture, a copy thereof is provided to the Noteholders, each Hedge Counterparty (if any), the Rating Agency and the Collateral Manager;

WHEREAS, the Issuer has determined (i) that this Supplemental Indenture is authorized and (ii) permitted under the Indenture and that the conditions set forth in the Indenture for entry into a supplemental indenture to implement the Benchmark Replacement Conforming Changes on <u>Appendix A</u> have been satisfied as of the date of the Supplemental Indenture; and

WHEREAS, the amendments set forth in this Supplemental Indenture shall take effect on June 30, 2023 (the "Amendment Effective Date");

NOW, THEREFORE, based upon the above recitals, the mutual premises and agreements contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned, intending to be legally bound, hereby agree as follows:

### Section 1. Amendments.

(a) The Indenture is hereby amended to delete the stricken text (indicated textually in the same manner as the following example: stricken text) and to add the bold and double-underlined text (indicated textually in the same manner as the following example: bold and double-underlined text) as set forth on the pages of the Indenture attached as Appendix A hereto. For the avoidance of doubt, the Floating Rate Notes will continue to accrue interest using the applicable LIBOR as the Benchmark for the remainder of the current Periodic Interest Accrual Period.

(b) The Notes and Exhibits to the Indenture are hereby amended (and deemed amended) as necessary in order to make such Notes and Exhibits consistent with the terms hereof by replacing references to "LIBOR" with "Benchmark" as applicable.

## Section 2. <u>Effect of Supplemental Indenture</u>.

- (a) Upon execution of this Supplemental Indenture, the Indenture shall be, and be deemed to be, modified and amended, effective as of the Amendment Effective Date, in accordance herewith and the respective rights, limitations, obligations, duties, liabilities and immunities of the Issuer and the Co-Issuer shall hereafter be determined, exercised and enforced subject in all respects to such modifications and amendments, and all the terms and conditions of this Supplemental Indenture shall be deemed to be part of the terms and conditions of the Indenture for any and all purposes. Except as modified and expressly amended by this Supplemental Indenture, the Indenture is in all respects ratified and confirmed, and all the terms, provisions and conditions thereof shall be and remain in full force and effect.
- (b) Except as expressly modified herein, the Indenture shall continue in full force and effect in accordance with its terms. All references in the Indenture to "the Indenture" or to "this Indenture" shall apply <u>mutatis mutandis</u> to the Indenture as modified by this Supplemental Indenture. The Trustee shall be entitled to all rights, protections, immunities and indemnities set forth in the Indenture as fully as if set forth in this Supplemental Indenture.

## Section 3. <u>Binding Effect</u>.

(a) The provisions of this Supplemental Indenture shall be binding upon and inure to the benefit of the Issuer, the Co-Issuer, the Collateral Manager, the Trustee, the Noteholders and each of their respective successors and assigns.

### Section 4. Acceptance by the Trustee.

(a) The Trustee accepts the amendments to the Indenture as set forth in this Supplemental Indenture and agrees to perform the duties of the Trustee upon the terms and conditions set forth herein and in the Indenture, subject to its protections, immunities and indemnities set forth therein and herein. Without limiting the generality of the foregoing, the Trustee assumes no responsibility for the correctness of the recitals contained herein, which shall be taken as the statements of the Co-Issuers and the Trustee shall not be responsible or accountable in any way whatsoever for or with respect to the validity, execution or sufficiency of this Supplemental Indenture and makes no representation with respect thereto.

### Section 5. Execution, Delivery and Validity.

(a) The Issuer and the Co-Issuer each represents and warrants to the Trustee that this Supplemental Indenture has been duly and validly executed and delivered by the Issuer or the Co-Issuer, as applicable, and constitutes its legal, valid and binding obligation, enforceable against the Issuer and the Co-Issuer, as applicable, in accordance with its terms.

### Section 6. GOVERNING LAW.

(a) THIS SUPPLEMENTAL INDENTURE SHALL BE CONSTRUED IN ACCORDANCE WITH, AND THIS SUPPLEMENTAL INDENTURE AND ANY MATTER ARISING OUT OF OR RELATING IN ANY WAY WHATSOEVER TO THIS SUPPLEMENTAL INDENTURE (WHETHER IN CONTRACT, TORT OR OTHERWISE), SHALL BE GOVERNED BY, THE LAW OF THE STATE OF NEW YORK.

## Section 7. <u>Counterparts</u>.

This Supplemental Indenture may be executed in several counterparts, each (a) of which shall be an original and all of which shall constitute but one and the same instrument. This Supplemental Indenture (and each related document, modification and waiver in respect of this Supplemental Indenture) may be executed and delivered in counterparts (including by facsimile or electronic transmission (including .pdf file, .jpeg file or any electronic signature complying with the U.S. federal ESIGN Act of 2000, including Orbit, Adobe Sign, DocuSign, or any other similar platform identified by the Issuer and reasonably available at no undue burden or expense to the Trustee)), each of which shall be deemed an original, and all of which together constitute one and the same instrument. Delivery of an executed counterpart signature page of this Supplemental Indenture by facsimile or any such electronic transmission shall be effective as delivery of a manually executed counterpart of this Supplemental Indenture and shall have the same legal validity and enforceability as a manually executed signature to the fullest extent permitted by applicable law. Any electronically signed document delivered via email from a person purporting to be an authorized officer shall be considered signed or executed by such authorized officer on behalf of the applicable person. The Trustee shall have no duty to inquire into or investigate the authenticity or authorization of any such electronic signature and shall be entitled to conclusively rely on any such electronic signature without any liability with respect thereto.

## Section 8. <u>Limited Recourse; Non-Petition</u>.

(a) The parties hereto agree to the provisions set forth in Sections 2.7(k) and 5.4(c) of the Indenture, and such provisions are incorporated in this Supplemental Indenture, mutatis mutandis.

### Section 9. Direction.

(a) By their signatures hereto, the Issuer and the Co-Issuer hereby direct the Trustee to execute this Supplemental Indenture, and acknowledge and agree that the Trustee will be fully protected in relying on the foregoing direction.

IN WITNESS WHEREOF, we have set our hands as of the day and year first written above.

**DRYDEN 87 CLO, LTD.,** 

as Issuer

By:

Name: Nicolas Rogivue

Title: Director

# **DRYDEN 87 CLO, LLC**, as Co-Issuer

# **DEUTSCHE BANK TRUST COMPANY AMERICAS**, as Trustee

Docusigned by: Sun-Hee Chang

Name: Sun-Hee Chang

Title: Assistant Vice President

DocuSigned by:

By: Name: Susan Gun

Title: VP

## ACKNOWLEDGED AND CONSENTED TO BY:

PGIM, INC.,

in its capacity as Collateral Manager

## Appendix A

DRYDEN 87 CLO, LTD., Issuer

DRYDEN 87 CLO, LLC, Co-Issuer

and

## **DEUTSCHE BANK TRUST COMPANY AMERICAS, Trustee**

**INDENTURE** 

Dated as of July 7, 2021

## TABLE OF CONTENTS

		<b>Page</b>
PRELIM	IINARY STATEMENT	1
GRANT	ING CLAUSES	1
1. I	DEFINITIONS	3
1.1	Definitions	
1.2	Rules of Construction	99
1.3	Assumptions as to Collateral Debt Obligations and Trust Estate	100
2	THE NOTES	102
2.1	Forms Generally	102
2.2	Form of Notes and Certificate of Authentication	103 <u>102</u>
2.3	Authorized Amount and Denominations	104
2.4	Execution, Authentication, Delivery and Dating	106
2.5	Registration, Registration of Transfer and Exchange	<u> 107</u> <u>106</u>
2.6	Mutilated, Destroyed, Lost or Stolen Notes	126
2.7	Payments on the Notes	127
2.8	Persons Deemed Owners	131
2.9	Purchase and Surrender of Notes; Cancellation	131
2.10	Definitive Notes	134
2.11	Determination of LIBOR; Benchmark Transition EventProvisions	135
2.12	Notes Beneficially Owned by Non-Permitted Holders	141
2.13	Tax Treatment and Tax Certification.	143
2.14	[Reserved]	<del>147</del> 146
2.15	[Reserved]	147 <u>146</u>
2.16	Additional Issuance	
2.17	Contributions.	
3.	CONDITIONS PRECEDENT AND SECURITY INTERESTS	152
3.1	General Provisions	
3.2	Security	154
3.3	Purchase of Initial Collateral Debt Obligations	155
3.4	Custodianship; Transfer of Collateral Debt Obligations and Eligible Investments	156

3.5	Collateral Debt Obligations Delivered After the Closing Date	157 <u>156</u>
3.6	Notes Issued After the Closing Date	157
4.	SATISFACTION AND DISCHARGE	<del>159</del> 158
4.1	Satisfaction and Discharge of Indenture	
4.2	Application of Trust Money	<u>161</u> 160
4.3	Repayment of Money Held by Paying Agent	161
5.	REMEDIES	161
5.1	Events of Default	161
5.2	Acceleration of Maturity; Rescission and Annulment	<u>164</u> <u>163</u>
5.3	Proceedings	<del>165</del> <u>164</u>
5.4	Remedies	165
5.5	Preservation of Trust Estate	<u>168</u> <u>167</u>
5.6	Trustee May File Proofs of Claim	169
5.7	Trustee May Enforce Claims Without Possession of Notes	<u>171</u> 170
5.8	Application of Money Collected	<del>171</del> <u>170</u>
5.9	Limitation on Suits	<u>172</u> 171
5.10	Unconditional Rights of Noteholders to Receive Principal and Interest.	172
5.11	Restoration of Rights and Remedies	<u>174</u> 173
5.12	Rights and Remedies Cumulative	<u>174</u> 173
5.13	Delay or Omission Not Waiver	174
5.14	Control by a Majority of the Controlling Class	174
5.15	Waiver of Past Defaults	
5.16	Undertaking for Costs	175
5.17	Waiver of Stay or Execution Laws	176 <u>175</u>
5.18	Sale of Trust Estate	
5.19	Action on Notes	
6.	THE TRUSTEE	177
6.1	Certain Duties and Responsibilities of the Trustee	177
6.2	Notice of Default	179
6.3	Certain Rights of Trustee	179
6.4	Not Responsible for Recitals or Issuance of Notes	183
6.5	May Hold Notes	183
6.6	Money Held in Trust	<del>184</del> <b>183</b>

	6.7	Compensation and Reimbursement	184
	6.8	Corporate Trustee Required; Eligibility	185
	6.9	Resignation and Removal of Trustee; Appointment of Successor	185
	6.10	Acceptance of Appointment by Successor Trustee	<u>187<mark>186</mark></u>
	6.11	Merger, Conversion, Consolidation or Succession to Business of Trus	tee187
	6.12	Certain Duties of Trustee Related to Delayed Payment of Proceeds	187
	6.13	[Reserved]	188
	6.14	Withholding	188
	6.15	Representative For Secured Noteholders Only; Agent For Each Other Party and the Holders of the Subordinated Notes	
	6.16	Authenticating Agents	189 <u>188</u>
	6.17	Assignment of Rights; Not Assumption of Duties	189
	6.18	Limitation on Duty of Trustee in Respect of the Trust Estate	<u>190<mark>189</mark></u>
	6.19	Representations and Warranties of The Bank	190
	6.20	Co-Trustees	191
7	7. R	EPRESENTATIONS AND COVENANTS	192
	7.1	Payment of Principal and Interest	192
	7.2	[Reserved]	<u>193<mark>192</mark></u>
	7.3	Maintenance of Office or Agency	<u>193<mark>192</mark></u>
	7.4	Money for Payments To Be Held in Trust	193
	7.5	Existence of Co-Issuers; Permitted Subsidiaries	<del>195</del> <u>194</u>
	7.6	Protection of Collateral	198
	7.7	Opinions and Other Documentation	199
	7.8	Performance of Obligations	199
	7.9	Negative Covenants	200
	7.10	Statement as to Compliance	<del>202</del> 201
	7.11	Co-Issuers May Not Consolidate or Merge	202
	7.12	No Other Business	202
	7.13	[Reserved]	<del>203</del> 202
	7.14	Calculation Agent	<del>203</del> 202
	7.15	Annual Rating Review; Notice of Rating	204
	7.16	Process Agent	204
	7.17	Additional Covenants	<u>205<mark>204</mark></u>
	7.18	Representations and Warranties of the Co-Issuers	<del>206</del> 205

7.19	Certain Tax Matters.	207
7.20	Maintenance of Listing	
7.21	Section 3(c)(7) Procedures	210 <u>209</u>
7.22	Certain Miscellaneous Covenants	211
7.23	Representations, Warranties and Undertakings of the Issuer	211
_	SUPPLEMENTAL INDENTURES	<del>214</del> <u><b>213</b></u>
8.1	Supplemental Indentures Without Consent of Noteholders	<u>214</u> 213
8.2	Supplemental Indentures With Consent of Noteholders	<del>220</del> 219
8.3	Execution of Supplemental Indentures	223 <mark>222</mark>
8.4	Effect of Supplemental Indentures	225 <mark>224</mark>
8.5	Reference in Notes to Supplemental Indentures	225 <u>224</u>
9. <u>I</u>	PRINCIPAL PREPAYMENTS; REDEMPTION OF NOTES	225
9.1	Principal Prepayment	
9.2	Notice to Trustee, Rating Agency and Collateral Manager	225
9.3	Notes Payable on Principal Prepayment Date	225
9.4	Optional Redemption; Election to Redeem	<del>226</del> 225
9.5	Notice by Issuer of Optional Redemption	229
9.6	Notice by Trustee of Optional Redemption or Clean-Up Call Red	demption230
9.7	Notes Payable on Redemption Date	231
9.8	Special Redemption	231
9.9	Ramp-Up Confirmation Failure	232
9.10	Clean-Up Call Redemption	
9.11	Optional Redemption of Secured Notes Using Additional Notes	or Secured
9.12	LoansRe-Pricing of Notes	
10. <i>I</i>	ACCOUNTS, ACCOUNTINGS AND RELEASES  Collection of Money	
10.1	Accounts; Collection Account; Unused Proceeds Account; Loan	
10.2	Account; Custodial Account; Interest Reserve Account; Permitte Account; Expense Reserve Account	ed Use
10.3	Release of Collateral Debt Obligations	
10.4	Reports by Trustee	252 <mark>251</mark>
10.5	Accountings	
10.6	Reports by Independent Accountants	263262

10.7	Reports to the Rating Agency	<del>264</del> 263
10.8	Inspection of Books and Records	264
11. A	PPLICATION OF MONEY	264
11.1	Disbursements of Money	264
12. <u>S</u>	ALE OF COLLATERAL DEBT OBLIGATIONS; SUBSTITUTION	271
12.1	Sale of Collateral Debt Obligations	271
12.2	Eligibility Criteria and Trading Restrictions	<del>275</del> <u>274</u>
12.3	[Reserved]	<del>280</del> 279
12.4	Collateral Debt Obligations Subject to Offer or Call; A/B Exchange	280 <mark>279</mark>
12.5	Purchase and Swap of Defaulted Obligations	281
12.6	[Reserved].	283
12.7	Conditions Applicable to All Transactions Involving Purchases of Colla Debt Obligations	
13. N	OTEHOLDERS' RELATIONS	<del>285</del> <b>28</b> 4
13.1	Subordination	
13.2	Standard of Conduct	285
14. M	MSCELLANEOUS  Forms of Decomposite Delivered to Trustee	<del>286</del> 285
14.1	Form of Documents Delivered to Trustee	286 <mark>285</mark>
14.2	Acts of Noteholders	286
14.3	Notices	287
14.4	Notices to Noteholders; Waiver; Communications with Rating Agency	
14.5	Effect of Headings and Table of Contents	294
14.6	Successors and Assigns	294
14.7	Severability	294
14.8	Benefits of Indenture	294
14.9	Governing Law	
14.10	Counterparts	
14.11	Jurisdiction	
14.12	Notices to S&P	
	Waiver of Jury Trial	
	Global Rating Condition	
14.14		
	Liability of Co-Issuers	297
14.15	Liability of Co-IssuersSSIGNMENT OF COLLATERAL MANAGEMENT AGREEMENT	

above) of all of the Classes of Notes and (e) when used with respect to the Subordinated Notes, the original principal amount thereof.

"<u>Alternative Reference Rate</u>" has the meaning specified in Section 2.11 hereof.

"Amendment Effective Date" means June 30, 3023.

"AML Compliance" means compliance with the Cayman AML Regulations.

"<u>AML Services Agreement</u>" means the agreement between the Issuer and the AML Services Provider (as amended from time to time) for the provision of services to the Issuer to enable the Issuer to achieve AML Compliance.

"AML Services Provider" means Maples Compliance Services (Cayman) Limited, a company incorporated in the Cayman Islands with its principal office at P.O. Box 1093, Boundary Hall, Cricket Square, Grand Cayman, KY1-1102, Cayman Islands.

"Applicable Issuer" or "Applicable Issuers" means, with respect to the Class A-1 Notes, the Class A-2 Notes, the Class B Notes, the Class C Notes and the Class D Notes, each of the Co-Issuers; with respect to the Class E Notes and the Subordinated Notes, the Issuer only; and with respect to any additional securities issued in accordance with Sections 2.16 and 3.6 hereof, the Issuer and, if such notes are co-issued, each of the Co-Issuers.

"Applicable Law" has the meaning specified in Section 6.3 hereof.

"Applicable Periodic Rate" means, with respect to each Class of Secured Notes, the <u>per annum</u> stated interest rate payable on such Class of Secured Notes set forth in the table below; <u>provided</u> that, if a Re-Pricing has occurred with respect to such Class of Secured Notes (if eligible for Re-Pricing), the Applicable Periodic Rate will be the applicable Re-Pricing Rate.

Class of Secured	Applicable Periodic Rate for each
<u>Notes</u>	Periodic Interest Accrual Period
Class A-1 Notes	Benchmark + 1.10% per annum
Class A-2 Notes	Benchmark + 1.35% per annum
Class B Notes	Benchmark + 1.55% per annum
Class C Notes	Benchmark + 1.90% per annum
Class D Notes	Benchmark + 2.95% per annum
Class E Notes	Benchmark + 6.15% per annum

"Applicable U.S. Risk Retention Regulations" means any U.S. law, rule or regulation which requires the Collateral Manager or an Affiliate thereof to purchase or retain credit risk in respect of securities issued by the Issuer.

"Approved Loan Index" means each of the following indexes, together with any other comparable nationally recognized loan index designated by the Collateral Manager upon written notice to the Rating Agency and the Collateral Administrator: CSFB Leveraged Loan Index, Deutsche Bank Leveraged Loan Index, Goldman Sachs/Loan Pricing Corporation Liquid ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023

DOC ID - 40325869.9

Leveraged Loan Index, Banc of America Securities Leveraged Loan Index, J.P. Morgan Leveraged Loan Indices and S&P/LSTA Leveraged Loan Index.

"Articles" means the Issuer's Memorandum of Association and Articles of Association, collectively.

"Asset-backed Commercial Paper" means commercial paper or other short-term obligations of a program that primarily issues externally rated commercial paper backed by assets or exposures held in a bankruptcy-remote, special purpose entity.

"<u>Asset Replacement Percentage</u>" has the meaning specified in Section 2.11 hereof.

"Assigned Moody's Rating" means the monitored publicly available facility rating or the monitored estimated rating expressly assigned to a Collateral Debt Obligation (or facility, as applicable) by Moody's that addresses the full amount of the principal and interest promised or a private letter rating.

"Assigned S&P Rating" means the monitored publicly available facility rating or the estimated rating expressly assigned to a Collateral Debt Obligation (or facility) by S&P that addresses the full amount of the principal and interest promised.

"Assumed Reinvestment Rate" means the greater of (a) LIBOR the applicable Benchmark (determined as of the applicable LIBOR Determination Datedetermination date) minus 0.25% per annum and (b) zero.

"<u>Authenticating Agent</u>" means, with respect to any Class of Notes, the Person appointed by the Trustee to authenticate such Notes on behalf of the Trustee pursuant to Section 6.16 hereof.

"Authorized Denomination" has the meaning specified in Section 2.3 hereof.

"Authorized Officer" means, with respect to the Issuer or the Co-Issuer, any director, chairman, deputy chairman, president, vice president, secretary, treasurer, members, partners, managers or other officer thereof or any chairman, deputy chairman, president, vice president, secretary, treasurer or other officer of any duly appointed agent thereof who is authorized to act for the Issuer or the Co-Issuer, as applicable, in matters relating to, and binding upon, the Issuer or the Co-Issuer, as applicable, and, for the avoidance of doubt, includes any duly appointed attorney-in-fact of such Issuer or the Co-Issuer. With respect to the Collateral Manager, any officer, employee or agent of the Collateral Manager who is authorized to act for the Collateral Manager in matters relating to, and binding upon, the Collateral Manager with respect to the subject matter of the request, certificate or order in question. With respect to the Trustee, a Responsible Officer. Each party shall be entitled to receive and accept a certification of the authority of any other party as conclusive evidence of the authority of any Person to act, and such certification may be considered as in full force and effect until receipt by such party of written notice to the contrary.

(c) if its payment obligations in respect of such Collateral Debt Obligation are guaranteed by a Person or entity (in a guarantee agreement with such person or entity, which guarantee agreement complies with S&P's then-current criteria with respect to guarantees) and the Collateral Manager elects to determine its Domicile pursuant to this clause (c) by written notice to the Collateral Administrator, the guarantor's country of organization.

"<u>DTC</u>" means the Depository Trust Company, its nominees and their respective successors.

"<u>Due Date</u>" means each date on which a distribution is due on a Pledged Obligation.

"Due Period" means, with respect to any Payment Date, the period beginning on and including the day following the last day of the immediately preceding Due Period (or, in the case of the Due Period that is applicable to the Initial Payment Date, beginning on the Closing Date) and ending (a) at the close of business on the last calendar day of the month immediately preceding the month in which such Payment Date occurs (or, for the last Due Period, ending at the close of business on the day preceding the Final Maturity Date) and (b) in the case of the final Due Period preceding an Optional Redemption in whole of the Secured Notes or a Clean-Up Call Redemption on a date that is not a Payment Date, on the Business Day preceding the Redemption Date. Amounts that would otherwise have been payable in respect of a Collateral Debt Obligation on the last day of a Due Period, but for such day not being a designated business day in the related underlying instruments of such Collateral Debt Obligation, shall be considered included in collections received during such Due Period.

"Effective Date Report" has the meaning specified in Section 9.9 hereof.

"Effective Spread" means, commencing with the first Periodic Interest Accrual Period after the Amendment Effective Date and continuing thereafter, with respect to any Floating Rate Collateral Debt Obligation (excluding Defaulted Obligations and only the portion of any Deferrable Collateral Debt Obligation which is currently deferring interest or paying such interest in kind), (i) if such Floating Rate Collateral Debt Obligation bears interest at a rate consisting of a spread plus a London interbank offered rate (a "LIBORan index based on SOFR (a "SOFR Obligation Rate"), the Effective Spread shall be the then-current per annum rate at which it pays interest in excess of the LIBOR Obligation RateSOFR based index in effect as of such time on such Floating Rate Collateral Debt Obligation; provided, that, if such Floating Rate Collateral Debt Obligation utilizes a minimum **LIBORSOFR** Obligation Rate (the "LIBORSOFR Floor") for the purposes of calculating interest due on such Floating Rate Collateral Debt Obligation (a "LIBORSOFR Floor Obligation") and the LIBORSOFR Floor is in effect, then such asset shall have an Effective Spread equivalent to the greater of (1) the spread of such asset, or (2) (a) the spread of such asset plus (b) the **LIBORSOFR** Floor minus (c) the then-current Benchmark rate in effect with respect to the Floating Rate Notes or (ii) if such Floating Rate Collateral Debt Obligation bears interest based on a non-London interbank offered rate based floating rate that is not an index based on SOFR, the Effective Spread shall be the then-current base rate applicable to such Floating Rate Collateral Debt Obligation (or, if such Floating Rate Collateral Debt Obligation utilizes a minimum interest base rate for the purposes of

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9

calculating interest due on such Floating Rate Collateral Debt Obligation (the "Floor Rate") and such Floor Rate is in effect, the then-current Floor Rate applicable to such Floating Rate Collateral Debt Obligation) <u>plus</u> the rate at which such Floating Rate Collateral Debt Obligation pays interest in excess of such base rate <u>minus</u> the then-current Benchmark rate in effect with respect to the Floating Rate Notes, which number may be less than zero. <u>Prior to the first Periodic Interest Accrual Period after the Amendment Effective Date</u>, the Effective Spread will be calculated in the same manner as prior to the Amendment Effective Date.

"Election to Retain" has the meaning set forth in Section 9.12(c).

"<u>Eligible EU/UK Retainer</u>" means an entity which qualifies as an "originator", "sponsor" or "original lender" (or other term having similar meaning to any of them) as defined in, and for purposes of compliance with, the EU/UK Securitization Requirements.

"Eligible Institution" means a corporation, association or trust company organized and doing business under the laws of the United States or of any state thereof, authorized under such laws to exercise corporate trust powers, having a combined capital and surplus of at least U.S.\$200,000,000 (or the equivalent in any other currency), being subject to supervision or examination by a Federal or state banking authority, and that (i) has a long-term issuer or credit rating of at least "BBB-" by S&P or a short-term issuer or credit rating of at least "A-3" by S&P (or any such lower rating which satisfies the S&P Rating Condition) and (ii) has an office within the United States.

"<u>Eligible Investment</u>" means any U.S. dollar-denominated investment that, at the time it is acquired by or delivered to the Trustee under this Indenture, is one or more of the following obligations or securities:

- (a) Cash;
- (b) (A) direct Registered obligations (1) of the United States of America or (2) the timely payment of principal of and interest on which is fully and expressly guaranteed by, the United States of America and (B) Registered obligations (1) of any agency or instrumentality of the United States of America the obligations of which are backed by the full faith and credit of the United States of America or (2) the timely payment of principal of and interest on which is fully and expressly guaranteed by such an agency or instrumentality, in each case if such agency or instrumentality has the Eligible Investment Required Ratings;
- (c) demand and time deposits in, and certificates of deposit of, bankers acceptances issued by, or federal funds sold by, (i) the Bank or its Affiliates or (ii) any other depository institution or trust company incorporated under the laws of the United States of America or any state thereof and subject to supervision and examination by federal and/or state banking authorities so long as, in each case of clause (i) and (ii) above, the commercial paper and/or other debt obligations of such depository institution or trust company at the

Exit) Regulations 2019 (SI 2019/660), in each case, together with any technical standards, guidance, implementing regulations or delegated regulations relating thereto and in each case as the same may be amended and/or superseded from time to time.

"<u>Euroclear</u>" means Euroclear Bank S.A./N.V., as operator of the Euroclear system.

"Euroclear Security" means a "security" (as defined in Section 8-102(a)(15) of the UCC) that (i) is a debt or equity security and (ii) is capable of being transferred to an Agent Member's account at Euroclear, whether or not such transfer has occurred.

"Event of Default" means any of the events of default set forth in Section 5.1 of this Indenture.

"Event of Default OC Test" has the meaning set forth in Section 5.1 hereof.

"Excel Default Model Input File" means an electronic spreadsheet file to be provided to S&P which file shall include the Balance in each Account and the following information (to the extent such information is available to the Collateral Manager and is not confidential) with respect to each Collateral Debt Obligation: (a) the name and country of Domicile of the issuer thereof and the particular issue held by the Issuer, (b) the CUSIP, the LoanX identifier (if any), or other applicable identification number associated with such Collateral Debt Obligation, (c) the par value of such Collateral Debt Obligation, (d) the type of issue (including, by way of example, whether such Collateral Debt Obligation is a bond, loan or asset-backed security), using such abbreviations as may be selected by the Trustee, (e) a description of the index or other applicable benchmark upon which the interest payable on such Collateral Debt Obligation is based (including, by way of example, fixed rate, step-up, zero coupon and LIBORSOFR), (f) the spread over the applicable index, (g) the S&P Industry Classification Group for such Collateral Debt Obligation, (h) the stated maturity date of such Collateral Debt Obligation, (i) the S&P Rating of such Collateral Debt Obligation or the issuer thereof, as applicable, (j) identification of Cov-Lite Loans, (k) the purchase price of any unsettled Collateral Debt Obligations and (1) identifying whether the Collateral Debt Obligation is a "first lien, last out" loan.

"Excepted Property" means (i) the U.S.\$250 transaction fee paid to the Issuer in consideration of the issuance of the Notes, (ii) the funds attributable to the issuance and allotment of the Issuer's Ordinary Shares or any bank account in the Cayman Islands in which such funds and the U.S.\$250 fee referred to in clause (i) of this definition are deposited (and any interest thereon), (iii) any Equity Security that is Margin Stock, (iv) all interest, including the membership interest, of the Issuer as the sole member of the limited liability company of the Co-Issuer and (v) the proceeds of any of the property described in clauses (i), (ii) and (iv) of this definition. Assets described in clauses (i), (ii) and (iv) above, and the proceeds thereof, are not available for distributions to Noteholders.

"Excess Par Amount" means an amount, as of any date of determination, equal to the greater of (a) zero and (b) the excess of (i) the sum of (A) the Aggregate Collateral Balance

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9 42

Custodian in the name of the Issuer, subject to the lien of the Trustee, for the benefit of the Secured Parties pursuant to Section 10.2(s) hereof.

"Fallback Rate" has the meaning specified in Section 2.11 hereof.

"FATCA" means Sections 1471 through 1474 of the Code, any current or future regulations or official interpretations thereof, any agreement entered into pursuant to Section 1471(b) of the Code, any intergovernmental agreement entered into in connection with the implementation of such Sections of the Code, or any U.S. or non-U.S. fiscal or regulatory legislation, rules, guidance notes or practices adopted pursuant to any such intergovernmental agreement.

"<u>Federal Reserve Bank of New York's Website</u>" has the meaning specified in Section 2.11 hereof.

"Fee Basis Amount" means an amount equal, for (a) the first Payment Date following the Closing Date, to the Aggregate Collateral Balance as of the last day of the related Due Period and (b) any other Payment Date (or other relevant date), to the Aggregate Collateral Balance on the first day of the related Due Period. With respect to any Collateral Management Fees payable on any Payment Date, the Fee Basis Amount that is calculated as of the beginning of the Due Period related thereto shall include the Balance of Eligible Investments in the Trust Estate purchased with Contributions.

"<u>Final Maturity Date</u>" means, with respect to any Class of Notes, the Stated Maturity Date with respect to such Class of Notes or such earlier date on which accrued but unpaid interest on (if applicable), and the Aggregate Principal Amount of, such Class is paid in full, including any such payment in full in connection with a Principal Prepayment, a Special Redemption, an Optional Redemption, a Clean-Up Call Redemption or a Ramp-Up Confirmation Failure.

"<u>Final Rule</u>" means the rule approved by the Federal Reserve Board, the Office of the Comptroller of the Currency, the Federal Deposit Insurance Corporation, the U.S. Securities and Exchange Commission and the U.S. Commodity Futures Trading Commission on June 25, 2020 amending the regulations implementing the Volcker Rule.

"Financial Asset" has the meaning specified in Section 8-102(a)(9) of the UCC.

"First LIBOR Determination End Date" means October 7, 2021.

"<u>First-Lien Last-Out Loan</u>" means a senior secured loan that, prior to a default with respect to such loan, is entitled to receive payments <u>pari passu</u> with other senior secured loans of the same obligor, but following a default becomes fully subordinated to other senior secured loans of the same obligor and is not entitled to any payments until such other senior secured loans are paid in full.

extent permitted in this Indenture or the Collateral Management Agreement, by an Authorized Officer of the Collateral Manager on behalf of the Issuer, or (ii) an order, request or direction provided in an e-mail by an Authorized Officer of the Issuer or the Co-Issuer, or by an Authorized Officer of the Collateral Manager on behalf of the Issuer, in each case except to the extent that the Trustee requests a written order; provided, however, that for the purposes of Section 10.3 and Article XII, and for the sale or acquisition of assets thereunder, "Issuer Order" or "Issuer Request" shall also mean the delivery to the Trustee on behalf of the Issuer, by email or otherwise, of a trade ticket, trade confirmation, instruction to trade or post (or similar language) which shall constitute direction and certification that the transaction is in compliance with the applicable prerequisites of Section 10.3 and Article XII, as the case may be.

"Junior Class" means, with respect to any specified Class of Notes, each Class of Notes that is subordinated to such Class, as indicated in Section 2.3 hereof.

"Junior Mezzanine Notes" has the meaning specified in Section 2.16(a) hereof.

"Letter of Credit" means a facility whereby (i) a fronting bank ("LOC Agent Bank") issues or will issue a letter of credit for or on behalf of a borrower pursuant to an Underlying Instrument, (ii) in the event that the letter of credit is drawn upon and the borrower does not reimburse the LOC Agent Bank, the lender/participant is obligated to fund its portion of the facility and (iii) the LOC Agent Bank passes on (in whole or in part) the fees and any other amounts it receives for providing the letter of credit to the lender/participant. The lender/participant may or may not be obligated to collateralize its funding obligations to the LOC Agent Bank.

"LIBOR" means the London interbank offered rate published on the Reuters Screen for U.S. dollar deposits (rounded to the nearest 0.00001%) for the applicable maturity as of the applicable determination date; provided that with respect to the interest rate of the Floating Rate Notes, LIBOR is determined as set forth in Section 2.11.

"LIBOR Banking Day" means a day on which commercial banks are open for business (including dealings in foreign currency deposits) in London.

"LIBOR Determination Date" has the meaning specified in Section 2.11(a) hereof.

"LIBOR Floor" has the meaning specified in the definition of "Effective Spread".

"LIBOR Floor Obligation" has the meaning specified in the definition of "Effective Spread".

"LIBOR Obligation Rate" means the London interbank offered rate.

"Lien" means any lien, mortgage, charge, encumbrance, adverse claim, security interest, hypothecation or other security device or arrangement of any kind or nature whatsoever.

"<u>Lifetime Zero Coupon Obligation</u>" means any obligation the terms of which provide for repayment of a stated principal amount at a stated maturity date but which do not ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9 51

applicable Collateral Coverage Test and on subsequent Payment Dates until paid in full (such applicable amount, the "Manager Contribution Repayment Amount"). For the avoidance of doubt, no rate of return or additional interest will accrue on any Manager Contribution.

"March 5, 2021 Announcements" means the March 5, 2021 announcements by ICE Benchmark Administration and the U.K. Financial Conduct Authority on future cessation and loss of representativeness of the LIBOR benchmarks.

"Margin Stock" has the meaning provided in Regulation U of the FRB.

"Market Value" means, with respect to any Collateral Debt Obligation, either:

- the product of the principal amount and the average of the bid price value (i) determined by the Loan Pricing Corporation, Mark-It Partners Inc., Interactive Data Corporation or any other nationally recognized pricing service that is Independent of the Collateral Manager and of which notice of the use thereof has been provided to S&P;
- (ii) if any such service is not available or applicable, then the product of the principal amount and the average of at least three firm bids obtained from dealers (that are Independent of the Collateral Manager and Independent of each other) that the Collateral Manager determines to be reasonably representative of the Collateral Debt Obligation's current market value and reasonably reflective of current market conditions:
- (iii) if only two such bids can be obtained, then the product of the principal amount and the lower of such two bids shall be the Market Value of the Collateral Debt Obligation;
- (iv) if only one such bid can be obtained, then the product of the principal amount and such bid shall be the Market Value of the Collateral Debt Obligation; and
- (v) if no bid described in subclauses (i) through (iv) can be obtained, then the Market Value of such Collateral Debt Obligation will be:
  - (1) first, the principal amount of such Collateral Debt Obligation multiplied by the lower of (x) the fair market value (expressed as a percentage of par) as determined by the Collateral Manager consistent with the procedures used by the Collateral Manager to determine the market value of an asset included in other funds managed by the Collateral Manager and (y) its S&P Recovery Rate; provided that, the Market Value of any such asset may not be determined in accordance with this clause (1) for a period longer than 30 days; and
  - (2) then, upon the expiration of the 30 day period specified in clause (1) above, the principal amount of such Collateral Debt Obligation multiplied by the least of (x) 70% of par, (y) its S&P Recovery Rate and (z) its fair market value (expressed as a percentage of par) as determined by the Collateral Manager

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 54

Retention Holder or (ii) has notified the Issuer that it has determined in good faith based on advice of nationally recognized counsel experienced in such matters that a proposed Refinancing or Re-Pricing would result in non-compliance or a material risk of non-compliance with any Risk Retention Regulation that is then applicable to the Retention Holder or its Affiliates or would impose new obligations on or increase the obligations of the Retention Holder or its Affiliates in complying with any such Risk Retention Regulation and (b) neither the Retention Holder nor an Affiliate of the Retention Holder is the Collateral Manager.

## "Reuters Screen" has the meaning set forth in Section 2.11 hereof.

"Revolving Loan Deposit" means, with respect to any Revolving Loan or Delayed Funding Loan, the deposit required to be delivered to the Trustee for deposit into the Loan Funding Account on the date on which such Revolving Loan or Delayed Funding Loan is acquired by the Issuer, in an amount equal to 100% of the Unfunded Commitment with respect thereto.

"Revolving Loans" means Loans that provide the borrower with a line of credit against which one or more borrowings may be made and that provide that such borrowed amounts may be repaid and re-borrowed from time to time (including funded and unfunded portions of revolving credit lines, unfunded commitments under specific facilities and other similar loans and investments); provided that any such Collateral Debt Obligation will be a Revolving Loan only until all commitments to make advances to the borrower expire or are terminated or irrevocably reduced to zero.

"Risk Retention Issuance" means an issuance of Notes to the Sponsor, the Retention Holder, a "majority-owned affiliate" (or other term having similar meaning, in each case, as defined in the U.S. Risk Retention Regulations or any other Risk Retention Regulation that is then applicable to the Collateral Manager or its Affiliates) of the Sponsor or an Eligible EU/UK Retainer as directed by the Issuer or the Retention Holder for the purpose of compliance with any Risk Retention Regulation that is then applicable to the Sponsor (if any), the Collateral Manager, the Retention Holder, an Eligible EU/UK Retainer or their respective Affiliates (including, without limitation, the U.S. Risk Retention Regulations or the EU/UK Securitization Requirements), as applicable, in connection with (i) a Refinancing, a Partial Redemption by Refinancing or a Re-Pricing (a "Refinancing Risk Retention Issuance"), or (ii) an issuance of additional Notes or Junior Mezzanine Notes pursuant to Section 2.16 (an "Additional Notes Risk Retention Issuance") (which may, for the avoidance of doubt, consist solely of an issuance of additional Notes or Junior Mezzanine Notes for the purpose of satisfying the Risk Retention Regulations). If, pursuant to Section 2.16, the Issuer is issuing additional Notes of an existing Class (or Junior Mezzanine Notes) to any other Person, the Additional Notes Risk Retention Issuance may consist of one or more Retention Holders purchasing a portion (which may be all) of such additional Notes (or Junior Mezzanine Notes) issued pursuant to Section 2.16. If the Issuer is issuing Secured Notes pursuant to a Refinancing, a Partial Redemption by Refinancing or a Re-Pricing, the Refinancing Risk Retention Issuance may consist of one or more Retention Holders purchasing a portion of the Secured Notes issued or amended in connection with such Refinancing, Partial Redemption by Refinancing or Re-Pricing.

"SOFR Floor" has the meaning specified in the definition of "Effective

Spread".

"SOFR Floor Obligation" has the meaning specified in the definition of "Effective Spread".

<u>"SOFR Obligation Rate"</u> has the meaning specified in the definition of "Effective Spread".

"Special Petition Expenses" has the meaning specified in Section 11.1(a)(i) hereof.

"Special Redemption" has the meaning specified in Section 9.8 hereof.

"Special Redemption Amount" has the meaning specified in Section 9.8 hereof.

"Special Redemption Date" has the meaning specified in Section 9.8 hereof.

"Specified Equity Securities" means (i) securities (including any Margin Stock but excluding any Loss Mitigation Loans) resulting from the exercise of an option, warrant, right of conversion, pre-emptive right, rights offering, credit bid or similar right in connection with the Restructuring of a Collateral Debt Obligation or (ii) an Equity Security received in connection with the Restructuring of a Collateral Debt Obligation; provided that the Investment Guidelines shall at all times apply with respect to Specified Equity Securities. The acquisition of Specified Equity Securities will not be required to satisfy the Reinvestment Criteria and Specified Equity Securities will not be included in the calculation of the Collateral Quality Tests or the Collateral Coverage Tests.

"Specified Equity Security Proceeds" means any proceeds received by the Issuer (including all Sale Proceeds and payments of interest and principal in respect thereof) on a Specified Equity Security of the type described in clause (i) of the definition of "Specified Equity Securities" acquired by the Issuer in accordance with Section 12.2(e).

"Sponsor" means, in relation to the transactions contemplated hereby, any "sponsor" (or other term having similar meaning, in each case as defined in and in accordance with the U.S. Risk Retention Regulations or any other Risk Retention Regulation that is then applicable to the Collateral Manager or its Affiliates).

"Stated Maturity Date" means May 20, 2034 (or, if such day is not a Business Day, the next succeeding Business Day).

"Step-Down Obligation" means an obligation or security which by the terms of the related Underlying Instruments provides for a decrease in the per annum interest rate on such obligation or security (other than by reason of any change in the applicable index or benchmark rate used to determine such interest rate) or in the spread over the applicable index or benchmark rate, solely as a function of the passage of time; *provided* that, an obligation or security providing

subject to the imposition of withholding tax (other than withholding tax on amendment fees, waiver fees, consent fees, extension fees, commitment fees or similar fees, to the extent such tax does not exceed 30% of such fees), which withholding is not compensated to the Issuer (in the case of a payment due to the Issuer), or is compensated by the Issuer (in the case of a payment due from the Issuer).

"<u>Tax Jurisdiction</u>" means (a) a sovereign jurisdiction that is commonly used as the place of organization of special purpose vehicles (including but not limited to the Bahamas, Bermuda, the British Virgin Islands, the U.S. Virgin Islands, Jersey, Singapore, the Cayman Islands, Ireland, the Channel Islands, St. Maarten and Curacao) and (b) any other jurisdiction that is commonly used as a place of organization of special purpose vehicles and is designated as a Tax Jurisdiction by the Collateral Manager from time to time.

"Tax Sensitive Equity Security" means any Equity Security acquired in connection with a workout or restructuring which, if held or received by the Issuer, could directly or indirectly (x) cause the Issuer to violate Section 7.19(a) of this Indenture, (y) cause the Issuer to be treated as engaged in a U.S. trade or business for U.S. federal income tax purposes or otherwise subject the Issuer to U.S. federal net income tax or (z) otherwise result in a material adverse tax consequence to the Issuer.

"Tax Sensitive Obligation" means collectively, (i) any Collateral Debt Obligation undergoing a workout or restructuring (including, for the avoidance of doubt, any Loss Mitigation Loan) which, if held or received by the Issuer, could directly or indirectly (x) cause the Issuer to violate Section 7.19(a) of this Indenture, (y) cause the Issuer to be treated as engaged in a U.S. trade or business for U.S. federal income tax purposes or subject the Issuer to net income tax in the United States or (z) result in a material adverse tax consequence to the Issuer, (ii) any Tax Sensitive Equity Security and (iii) any other asset owned by the Issuer, if the Issuer discovers that its ownership of such asset could cause the Issuer to be treated as engaged in a U.S. trade or business for U.S. federal income tax purposes or otherwise result in a material adverse tax consequence to the Issuer.

"Term SOFR" has the meaning specified in Section 2.11 hereof.

"Term SOFR Administrator" has the meaning specified in Section 2.11

hereof.

"Term SOFR Reference Rate" has the meaning specified in Section 2.11

hereof.

"<u>Tested Items</u>" has the meaning specified in Section 9.9 hereof.

"Total Diversity Score" means a single number that measures concentrations among the Collateral Debt Obligations in the Trust Estate, in terms of both obligors and obligor industries, in the manner set forth below. As of any date of determination, the Total Diversity Score for the Collateral Debt Obligations in the Trust Estate (other than Defaulted Obligations and Equity Securities) is the sum of the Industry Diversity Scores for all Moody's Industry

Class	Class A-1 Notes	Class A-2 Notes	Class B Notes	Class C Notes	Class D Notes	Class E Notes	Subordinated Notes
Applicabl e Periodic Rate (per annum) <sup>1</sup>	Benchmark + 1.10%	Benchmark + 1.35%	Benchmark + 1.55%	Benchmark + 1.90%	Benchmark + 2.95%	Benchmark + 6.15%	N/A
Stated Maturity Date <sup>2</sup>	May 20, 2034	May 20, 2034	May 20, 2034	May 20, 2034	May 20, 2034	May 20, 2034	May 20, 2034
Priority Classes	None	A-1	A-1, A-2	A-1, A-2, B	A-1, A-2, B, C	A-1, A-2, B, C, D	A-1, A-2, B, C, D, E
Pari Passu Classes	None	None	None	None	None	None	None
Junior Classes	A-2, B, C, D, E, Subordinated Notes	B, C, D, E, Subordinated Notes	C, D, E, Subordinated Notes	D, E, Subordinated Notes	E, Subordinated Notes	Subordinate d Notes	None

Following a Re-Pricing of any Class of Re-Pricing Eligible Secured Notes, the Applicable Periodic Rate for such Class will be the Benchmark plus the Re-Pricing Rate for such Class. The Benchmark for the Periodic Interest Accrual Period in effect as of the Amendment Effective Date shall remain in effect until the end of such Periodic Interest Accrual Period. For Periodic Interest Accrual Periods beginning after the Amendment Effective Date, the Benchmark shall be determined in accordance withas provided in Section 2.11.

Each Class of Notes, whether issued in the form of Definitive Notes or Global Notes, shall be issuable as of the Closing Date in minimum authorized denominations of (i) in the case of the Class A-2 Notes, the Class B Notes, the Class C Notes, the Class E Notes and the Subordinated Notes, U.S.\$250,000 and (ii) in the case of the Class A-1 Notes and the Class D Notes, U.S.\$100,000, and, in each case, integral multiples of U.S.\$1.00 in excess thereof, unless the Issuer otherwise consents. The minimum denominations of the Notes authorized to be issued under this Section 2.3 are referred to herein in each case as an "Authorized Denomination" and are expressed in terms of the principal amounts thereof at the date of issuance. After issuance, any Note may fail to be in an Authorized Denomination due to the repayment of principal thereof in accordance with the Priority of Payments or any other applicable provision hereof, and after such repayment the "Authorized Denomination" of any such Note, for purposes of this Indenture, shall mean the original Authorized Denomination reduced by any such repayment.

### 2.4 Execution, Authentication, Delivery and Dating.

The Notes shall be executed on behalf of the Applicable Issuers by an Authorized Officer or Authorized Officers thereof. The signature or signatures of such Authorized Officers on the Notes may be manual, facsimile or electronic.

The Stated Maturity Date shall be such date or, if such date is not a Business Day, the next following Business Day.

certificates in exchange for the applicable Global Notes to the beneficial owners thereof; and in the case of clauses (iii) or (iv) of this Section 2.10(a), the Issuer will issue or cause to be issued a Definitive Note in exchange for the applicable Global Notes (or interest therein) to the transferee thereof.

- (b) Any Global Note that is transferable in the form of a Definitive Note to the beneficial owners thereof pursuant to this Section 2.10 shall be surrendered by the Depository to the Trustee's office located in Jacksonville, Florida (as stated in the definition of "Corporate Trust Office") to be so transferred, in whole or from time to time in part, without charge, and the Applicable Issuers shall execute and the Trustee shall authenticate and deliver, upon such transfer of each portion of such Global Note, an equal Aggregate Principal Amount of definitive physical certificates (pursuant to the instructions of the Depository) in Authorized Denominations. Any interest in a Global Note delivered in exchange for a Definitive Note shall, except as otherwise provided by Section 2.5(j) hereof, bear the legends set forth in the applicable Exhibit hereto and shall be subject to the transfer restrictions referred to in such legends.
- (c) Subject to the provisions of clause (b) of this Section 2.10, the Holder of a Global Note may grant proxies and otherwise authorize any Person, including Agent Members and Persons that may hold interests through Agent Members, to take any action which such Holder is entitled to take under this Indenture or such Notes.
- (d) In the event of the occurrence of any of the events specified in subclauses (i), (ii) and (iii) of subsection (a) of this Section 2.10, the Applicable Issuers will promptly make available to the Trustee a reasonable supply of Definitive Notes in definitive, fully registered form without interest coupons. The Definitive Notes shall be in substantially the same form as the Exhibits to this Indenture with such changes therein as the Applicable Issuers and Trustee shall agree, and the Applicable Issuers shall execute, and the Trustee shall authenticate and deliver, in exchange for the Global Note or Global Notes, as the case may be, the same Aggregate Principal Amount of Definitive Notes of Authorized Denominations.

#### 2.11 Determination of LIBOR: Benchmark Transition Event Provisions.

(a) For each Periodic Interest Accrual Period, the Benchmark for the Floating Rate Notes will be determined by the Calculation Agent (and in each case rounded to the nearest 0.00001%) in accordance with the following provisions (as provided in this Indenture (including Section 7.14); provided that, in respect of the Floating Rate Notes, such rate will not be less than 0%):.

(a) If the Benchmark is LIBOR, on the second Business Day (provided such day is also a LIBOR Banking Day, and otherwise on the next preceding Business Day that is also a LIBOR Banking Day) prior to the commencement of such Periodic Interest Accrual Period (and, in the case of the first Periodic Interest Accrual Period following the Closing Date, the second Business Day prior to the Closing Date or the First LIBOR Determination End Date, as applicable) (each such day, a "LIBOR Determination Date"), LIBOR will equal the rate for deposits with a term of three months, as obtained by the Calculation Agent by reference to Reuters Page LIBOR01 or such other page as may replace such Reuters Page LIBOR01, as of

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9

11:00 a.m. (London time) on such LIBOR Determination Date; provided, that LIBOR (x) for the period from the Closing Date to the First LIBOR Determination End Date will equal the rate appearing on the Reuters Screen, or such other page as may replace such Reuters Screen, for deposits with a term of three months, and (y) will be reset on the second Business Day prior to the First LIBOR Determination End Date for the remaining portion of the first Periodic Interest Accrual Period by interpolating based on the rate for the next shorter period of time for which rates are available and the rate for the next longer period of time for which rates are available.

(b) If, on any LIBOR Determination Date, such rate does not appear on Reuters Page LIBOR01 or such other page as may replace such Reuters Page LIBOR01 (the "Reuters Screen"), LIBOR for the immediately following Periodic Interest Accrual Period shall be LIBOR as determined on the previous LIBOR Determination Date; provided, however, that, (i) if LIBOR is not published on the Reuters Screen on a LIBOR Determination Date and the Collateral Manager has determined that a Benchmark Transition Event and its related Benchmark Replacement Date has not occurred, and (ii) LIBOR for such LIBOR Determination Date thereafter appears on the Reuters Screen within 5 Business Days following such LIBOR Determination Date, then LIBOR for the immediately following Periodic Interest Accrual Period will be LIBOR as published on the Reuters Screen. Until such time as LIBOR is published on the Reuters Screen, LIBOR will be provisionally reported on the applicable Note Valuation Report as "subject to publication" until it is published on the Reuters Screen (and the Note Valuation Report will thereafter be updated to include LIBOR as published on the Reuters Screen and made available by the Trustee to authorized recipients of the Note Valuation Report).

(b) Gerenchmark Replacement. If the Collateral Manager determines that a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of any determination of the then-current Benchmark on any date, upon written notice from the Collateral Manager to the Issuer, the Calculation Agent, the Collateral Administrator and the Trustee (who will forward such notice to the Holders and the Rating Agency), the Alternative Reference Rate will replace the then-current Benchmark for all purposes relating to the Floating Rate Notes in respect of such determination on such date and all determinations on all subsequent dates. A supplemental indenture shall not be required in order to adopt an Alternative Reference Rate.

with the implementation of an Alternative Reference Rate, the Issuer, the Co-Issuer and the Trustee shall execute one or more supplemental indentures to implement Benchmark Replacement Conforming Changes from time to time, pursuant to Section 8.1(a)(xxxvii), and such supplemental indentures will become effective without any further action or consent of any Noteholder or any other Person. The Trustee's and Calculation Agent's responsibility and liability in connection with the foregoing are subject to the limitations set forth in Section 7.14(c).

(d) (e) Decisions and Determinations. Any determination, decision or election that may be made by the Collateral Manager pursuant to this Section 2.11, including any determination with respect to a tenor or maturity, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error, may be made in ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023

DOC ID - 40325869.9 140

the Collateral Manager's sole discretion, and, notwithstanding anything to the contrary in the documentation relating to the Notes or the Issuer, shall become effective without consent from any other party.

have any liability for any interest rate published by any publication that is the source for determining the interest rates of the Floating Rate Notes, including but not limited to the Reuters Sereen (or any successor source), rates compiled by the ICE rates published by the CME Group Benchmark Administration Ltd.Limited (CBA) or any successor thereto, or rates published by the FRB or on the Federal Reserve Bank of New York's Website. The Collateral Manager shall have no liability to the Issuer, the Co-Issuer, the Trustee, any Holder or beneficial owner of Notes or any other Person for the determination or selection of (or any failure to determine or select) any Alternative Reference Rate or Benchmark Replacement (including any modifier thereto), or for the determination of (or failure to determine) whether the conditions to the designation of such Alternative Reference Rate or Benchmark Replacement, including a Benchmark Transition Event or Benchmark Replacement Date, have occurred or the date of such occurrence, or for the adoption of (or failure to adopt) a supplemental indenture for Benchmark Replacement Conforming Changes or for any other purpose.

(f) (g) Certain Defined Terms. As used in this Section 2.11:

"Alternative Reference Rate" means a replacement rate for the <a href="then-current">then-current</a>
Benchmark that is the Benchmark Replacement (as determined by the Collateral Manager with written notice to the Issuer, the Trustee (who shall forward notice to the Holders of the Secured Notes and the Holders of the Subordinated Notes), the Collateral Administrator and the Calculation Agent). If the Benchmark Replacement cannot be determined by the Collateral Manager as of the Benchmark Replacement Date, then the Alternative Reference Rate shall mean the first alternative set forth in the order below that the Collateral Manager determines can be determined as of the Benchmark Replacement Date: (1) the rate proposed by the Collateral Manager and consented to by a Majority of the Controlling Class and a Majority of the Subordinated Notes, and (2) the Fallback Rate; <a href="provided">provided</a>, that the Alternative Reference Rate for the Floating Rate Notes will be no less than zero. Notice of any such determination shall be delivered to the Issuer, the Trustee (who shall forward notice to the Holders of the Secured Notes and the Holders of the Subordinated Notes), the Collateral Administrator and the Calculation Agent.

"Asset Replacement Percentage" means, on any date of calculation, a fraction (expressed as a percentage) where the numerator is the outstanding principal balance of the Collateral Debt Obligations that were are indexed to the a reference rate identified in the definition of "Benchmark Replacement" for the Index Maturity as a potential replacement for the then-current Benchmark as of such calculation date and the denominator is the outstanding principal balance of the Collateral Debt Obligations as of such calculation date.

"Benchmark" means, initially, LIBOR the sum of (i) Term SOFR plus (ii) 0.26161%; provided that if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred with respect to LIBOR Term SOFR or the then-current

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9

Benchmark, then "Benchmark" means the applicable Alternative Reference Rate; <u>provided</u> <u>further</u> that, in any event, the Benchmark will not be less than 0%.

"Benchmark Replacement" means the first alternative set forth in the order below, if any, that the Collateral Manager determines can be determined as of the Benchmark Replacement Date, in each case only if such rate is being used by at least 50% of the Aggregate Principal Amount of the Floating Rate Collateral Debt Obligations included in the Trust Estate that pay interest quarterly:

- (1) the sum of (a) Term SOFR and (b) the Benchmark Replacement Adjustment[Reserved];
- (2) the sum of (a) Daily Simple SOFR and (b) the applicable Benchmark Replacement Adjustment;
- (3) the sum of (a) the alternate rate of interest that has been selected or recommended by the Relevant Governmental Body as the replacement for the then-current Benchmark for the Index Maturity and (b) the Benchmark Replacement Adjustment;
- (4) the sum of (a) the ISDA Fallback Rate and (b) the Benchmark Replacement Adjustment; and
- (5) the sum of: (a) the alternate rate of interest that has been selected by the Collateral Manager as the replacement for the then-current Benchmark for the Index Maturity giving due consideration to any industry-accepted rate of interest as a replacement for the then-current Benchmark for similar Dollar-denominated collateralized loan obligation securitization transactions at such time and (b) the Benchmark Replacement Adjustment;

provided, that, (A) if a Benchmark Replacement is selected pursuant to clause (2) above, then on the first day the Collateral Manager determines that a redetermination of the Benchmark Replacement on such date would result in the selection of a Benchmark Replacement under clause (1) above, then (x) the Benchmark Replacement Adjustment shall be redetermined on such date utilizing the Unadjusted Benchmark Replacement corresponding to the Benchmark Replacement under clause (1) above and (y) such redetermined Benchmark Replacement shall become the Benchmark on each determination date on or after such date; and (B) if redetermination of the Benchmark Replacement on such date as described in the preceding clause (A) would not result in the selection of a Benchmark Replacement under clause (1) above, then the Benchmark shall remain the Benchmark Replacement as previously determined pursuant to clause (2) above.

"Benchmark Replacement Adjustment" means the first alternative set forth in the order below that can be determined by the Collateral Manager as of the Benchmark Replacement Date:

(1) the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected, endorsed

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (1) a public statement or publication of information by or on behalf of the administrator of the Benchmark announcing that the administrator has ceased or will cease to provide the Benchmark permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide the Benchmark;
- (2) a public statement or publication of information by the regulatory supervisor for the administrator of the Benchmark, the central bank for the currency of the Benchmark, an insolvency official with jurisdiction over the administrator for the Benchmark, a resolution authority with jurisdiction over the administrator for the Benchmark or a court or an entity with similar insolvency or resolution authority over the administrator for the Benchmark, which states that the administrator of the Benchmark has ceased or will cease to provide the Benchmark permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide the Benchmark;
- (3) a public statement or publication of information by the regulatory supervisor for the administrator of the Benchmark announcing that the Benchmark is no longer representative; or
  - (4) the Asset Replacement Percentage exceeds 50%.

For the avoidance of doubt, although the March 5, 2021 Announcements constitute a Benchmark Transition Event, such event has not caused the occurrence of a Benchmark Replacement Date as of the Closing Date.

"Daily Simple SOFR" means, for any day, SOFR, with the conventions for this rate (which will include a lookback) being established by the Collateral Manager in accordance with the conventions for this rate selected or recommended by the Relevant Governmental Body for determining "Daily Simple SOFR" for leveraged loans; provided, that if the Collateral Manager decides (in its sole discretion) that any such convention is not administratively feasible for the Collateral Manager, then the Collateral Manager may establish another convention in its reasonable discretion.

"Fallback Rate" means the sum of (1) the Reference Rate Modifier and (2) as determined by the Collateral Manager in its commercially reasonable discretion, either (x) the quarterly pay reference rate recognized or acknowledged as being the industry standard replacement rate for leveraged loans (which recognition may be in the form of a press release, a member announcement, member advice, letter, protocol, publication of standard terms or otherwise) by the Loan Syndications and Trading Association or the Relevant Governmental Body, (y) the quarterly pay reference rate that is used in calculating the interest rate of at least 50% of the Collateral Debt Obligations (by par amount), as determined by the Collateral Manager as of the first day of the Periodic Interest Accrual Period during which such determination is made or (z) the rate that is consistent with the reference rate being used with respect to at least 50% (by principal amount) of the floating rate securities issued in the

new-issue collateralized loan obligation market and/or floating rate securities in collateralized loan obligation market that have amended their reference rate, in each case in the preceding three months from the date of determination that bear interest based on a base rate other than the LIBOR Term SOFR; provided, that if the Collateral Manager determines that a Benchmark Replacement becomes determinable at any time when the Fallback Rate is effective, then such Benchmark Replacement shall become the Benchmark as determined in accordance with the procedures of this Section 2.11.

"<u>Federal Reserve Bank of New York's Website</u>" means the website of the Federal Reserve Bank of New York at http://www.newyorkfed.org, or any successor source.

"Index Maturity" means a term of three months.

"ISDA Definitions" means the 2006 ISDA Definitions published by the International Swaps and Derivatives Association, Inc. or any successor thereto, as amended or supplemented from time to time, or any successor definitional booklet for interest rate derivatives published from time to time.

"ISDA Fallback Adjustment" means the spread adjustment, (which may be a positive or negative value or zero) that would apply for derivatives transactions referencing the ISDA Definitions to be determined upon the occurrence of an index cessation event with respect to the Benchmark for the applicable tenor.

"ISDA Fallback Rate" means the rate that would apply for derivatives transactions referencing the ISDA Definitions to be effective upon the occurrence of an index cessation date with respect to the Benchmark for the applicable tenor excluding the applicable ISDA Fallback Adjustment.

"Reference Rate Modifier" means a modifier, other than the Benchmark Replacement Adjustment, recognized or acknowledged by the Loan Syndication and Trading Association or the Alternative Reference Rates Committee convened by the Federal Reserve that is applied to a reference rate to the extent necessary to cause such rate to be comparable to the three-month LIBORthen-current Benchmark, which may include an addition to or subtraction from such unadjusted rate.

"Reference Time" with respect to any determination of the Benchmark means (1) if the Benchmark is LIBOR, 11:00 a.m. (London based on Term SOFR, 6:00 a.m. (New York City time) on the day that is two London banking days U.S. Government Securities Business Days preceding the date of such determination, and (2) if the Benchmark is not LIBOR based on Term SOFR, the time determined in accordance with the Benchmark Replacement Conforming Changes.

"Relevant Governmental Body" means the Federal Reserve Board and/or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Federal Reserve Board and/or the Federal Reserve Bank of New York or any successor thereto.

"SOFR" with respect to any day means the secured overnight financing rate published for such day by the Federal Reserve Bank of New York, as the administrator of the benchmark, (or a successor administrator) on the Federal Reserve Bank of New York's Website.

"SOFR Determination Date" means, with respect to each Periodic Interest Accrual Period, the second U.S. Government Securities Business Day preceding the first day of such Periodic Interest Accrual Period.

"Term SOFR" means, with respect to the Notes for any Periodic Interest
Accrual Period, the Term SOFR Reference Rate for the Index Maturity, as such rate is
published by the Term SOFR Administrator at 6:00 a.m. (New York City time) on the
SOFR Determination Date with respect to such Periodic Interest Accrual Period; provided
that if as of 5:00 p.m. (New York City time) on any SOFR Determination Date, the Term
SOFR Reference Rate for the Index Maturity has not been published by the Term SOFR
Administrator, then Term SOFR shall be (x) the Term SOFR Reference Rate for the Index
Maturity as published by the Term SOFR Administrator on the first preceding U.S.
Government Securities Business Day for which such Term SOFR Reference Rate for the
Index Maturity was published by the Term SOFR Administrator, so long as such first
preceding U.S. Government Securities Business Day is not more than five Business Days
prior to such SOFR Determination Date or (y) if the Term SOFR Reference Rate cannot be
determined in accordance with clause (x) of this proviso, Term SOFR shall be the Term
SOFR Reference Rate as determined on the previous SOFR Determination Date.

"Term SOFR Administrator" means CME Group Benchmark
Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference
Rate that the Collateral Manager has selected in its reasonable discretion and of which the
Collateral Manager has provided written notice to the Trustee and the Collateral
Administrator).

"<u>Term SOFR Reference Rate</u>" means thea forward-looking term rate for the Index Maturity based on SOFR that has been selected or recommended by the Relevant Governmental Body.

"<u>Unadjusted Benchmark Replacement</u>" means the Benchmark Replacement excluding the applicable Benchmark Replacement Adjustment.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

### 2.12 Notes Beneficially Owned by Non-Permitted Holders.

(a) Notwithstanding anything to the contrary elsewhere in this Indenture, any transfer of a beneficial interest in any Rule 144A Global Notes or Rule 144A Definitive Note to a Person that is not both (i) a Qualified Purchaser and (ii) a Qualified

ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023 DOC ID - 40325869.9

its assets to any other Person, with the exception of sales and exchanges of Pledged Obligations contemplated hereunder.

#### 7.12 No Other Business.

The Issuer shall have no employees (other than its directors) and shall not engage in any business or activity other than (i) issuance, payment and redemption of the Ordinary Shares, (ii) issuance, payment, redemption, re-pricing and refinancing of the Secured Notes, (iii) issuance, payment and redemption of the Subordinated Notes, (iv) issuance, payment, redemption, re-pricing and refinancing of any additional notes issued pursuant to this Indenture, (v) acquiring, owning, managing, holding, pledging and selling solely for its own account Collateral Debt Obligations and Eligible Investments and any other instrument or property included in the Trust Estate as permitted by this Indenture, (vi) the execution and delivery of, and performance under, the Transaction Documents, (vii) admission as the sole member of the Co-Issuer and ownership of, directly or indirectly, the equity interests in any Permitted Subsidiary, (viii) entering into Hedge Agreements for its own account and (ix) other activities incidental or necessary to the foregoing and permitted by this Indenture.

The Issuer shall not hold itself out as originating loans, lending funds, making a market in loans or other assets, selling loans or other assets to customers or as willing to enter into, assume, offset, assign or otherwise terminate positions in derivative financial instruments with customers.

The Co-Issuer shall have no employees and shall not engage in any business or activity other than (i) the co-issuance of the Co-Issued Notes and any additional notes co-issued pursuant to this Indenture, (ii) redemption of its equity capital (if any) and (iii) engaging in any other incidental activities. The Co-Issuer will not have any claim on the Collateral. The Issuer and the Co-Issuer shall amend their organizational documents only if the Global Rating Condition with respect to any Class of Notes then rated by the Rating Agency is satisfied.

## 7.13 [Reserved]

### 7.14 Calculation Agent.

Notes remain Outstanding there will at all times be an agent appointed (which does not control, and is not controlled by or under common control with, the Issuer or its Affiliates or the Collateral Manager or its Affiliates) to calculate the Benchmark in respect of each Periodic Interest Accrual Period in accordance with the terms of Section 2.11 hereof (the "Calculation Agent"). The Co-Issuers hereby appoint the Collateral Administrator as Calculation Agent for purposes of determining the Benchmark for each Periodic Interest Accrual Period. The Calculation Agent may be removed by the Co-Issuers at any time. If the Calculation Agent is unable or unwilling to act as such or is removed by the Co-Issuers, the Co-Issuers will promptly appoint as a replacement Calculation Agent a leading bank which is engaged in transactions in Eurodollar deposits in the international Eurodollar market and which does not control and is not

controlled by or under common control with the Co-Issuers or their Affiliates. The Calculation Agent may not resign its duties without a successor having been duly appointed.

The Calculation Agent hereby agrees that, so long as the Benchmark is LIBOR, as soon as possible after 11:00 a.m. (Londonbased on Term SOFR, as soon as practicable after 6:00 a.m. (New York time) on each LIBORSOFR Determination Date, but in no event later than 11:00 a.m. (London New York time) on the U.S. Government Securities Business Day immediately following each **LIBORSOFR** Determination Date, the Calculation Agent will calculate the Applicable Periodic Rate for the next Periodic Interest Accrual Period and the amount of interest for such Periodic Interest Accrual Period payable on the related Payment Date in respect of each U.S.\$1,000,000 principal amount of the Floating Rate Notes of each Class (rounded to the nearest cent, with half a cent being rounded upward), and the amount of interest payable for such Periodic Interest Accrual Period payable on the related Payment Date in respect of Fixed Rate Notes (if any) of each Class (rounded to the nearest cent, with half a cent being rounded upward), and will communicate such rates and amounts to the Co-Issuers, the Trustee, each Paying Agent, the Depository, Euroclear, Clearstream and the Collateral Manager. The Calculation Agent will also specify to the Co-Issuers the quotations upon which the Applicable Periodic Rate with respect to each Class of Floating Rate Notes is based, and in any event the Calculation Agent shall notify the Issuer before 7:00 p.m. (London New York time) on each LIBOR SOFR Determination Date that either: (i) it has determined or is in the process of determining the Applicable Periodic Rate and the applicable amount of Periodic Interest with respect to each Class of Floating Rate Notes or (ii)if it has not determined and is not in the process of determining the Applicable Periodic Rate and the amount of Periodic Interest for such Floating Rate Notes, together with its reasons therefor.

The determination of the Applicable Periodic Rate and the amount of Periodic Interest with respect to each Class of Secured Notes by the Calculation Agent shall, in the absence of manifest error, be final and binding upon all parties.

shall be under any obligation (i) to monitor, determine or verify the unavailability or cessation of LIBOR (or other applicablethe Benchmark), or whether or when there has occurred, or to give notice to any other Transaction Party of the occurrence of, any Benchmark Transition Event or Benchmark Replacement Date, (ii) to select, determine or designate any Alternative Reference Rate, Benchmark Replacement or Fallback Rate, or other successor or replacement benchmark index, or whether any conditions to the designation of such a rate have been satisfied, or (iii) to select, determine or designate any Benchmark Replacement Adjustment, Reference Rate Modifier, or other modifier to any replacement or successor index, or (iv) to determine whether or what Benchmark Replacement Conforming Changes are necessary or advisable, if any, in connection with any of the foregoing. This section shall not affect the obligation of the Calculation Agent to calculate the replacement Benchmark in accordance with the then-current procedures set forth in this Indenture.

(d) None of the Trustee, the Paying Agent or the Calculation Agent shall be liable for any inability, failure or delay on its part to perform any of its duties set forth in this Indenture as a result of the unavailability of LIBOR (or other the applicable Benchmark) and ChangePro Comparison of 40325869v1 and 40325869v9 06/27/2023

DOC ID - 40325869.9 211

Calculation Date for the second Payment Date following the Closing Date the Issuer shall cause the Collateral Administrator to compile and make available to the Rating Agency the Effective Date Report. Notwithstanding the foregoing, if (i) the Issuer shall have satisfied the Target Initial Par Condition, (ii) the Issuer shall have caused the Collateral Administrator to compile and make available to the Rating Agency a report (the "Effective Date Report"), determined as of the Ramp-Up End Date, containing the following items: (A) the information reported in a Monthly Report under Section 10.5(a), other than information described in Section 10.5(a) which is not applicable as of the date of such report or not required by a Rating Agency to which the Effective Date Report is delivered, and (B) whether the Target Initial Par Condition is satisfied and (iii) the Issuer shall have caused its accountants appointed pursuant to Section 10.6 to provide to the Trustee a report that applies agreed-upon procedures and specifies the procedures applied (the "Accountants' Report"), re-calculating and comparing the following items in the Effective Date Report: (1) with respect to each Collateral Debt Obligation, by reference to such sources as shall be specified therein, the obligor name, coupon/spread, maturity date, principal balance, and S&P Rating, (2) as of the Ramp-Up End Date, each of the Principal Coverage Tests, the Collateral Quality Tests (excluding the S&P CDO Monitor Test) and the Percentage Limitations and (3) the Target Initial Par Condition (the foregoing clauses (2) and (3) are collectively the "Tested Items"). For the avoidance of doubt, the Effective Date Report shall not include or refer to the Accountants' Report (except to the extent required by applicable law or regulation or by any governmental or regulatory body).

If (1) the S&P CDO Model Election Date has not yet occurred as of the last day of the Ramp-Up Period, (2) the Issuer provides (or causes the Collateral Administrator to provide) S&P with the Excel Default Model Input File, (3) the Issuer causes the Collateral Administrator or the Collateral Manager to make available to S&P the Effective Date Report and (4) an Effective Date Report run as of the last day of the Ramp-Up Period is provided to S&P that also includes an indication that the S&P CDO Monitor Test, as adjusted by the S&P CDO Monitor Formula Adjustments, result is passing, then a written confirmation from S&P of its Initial Ratings of the Secured Notes shall be deemed to have been provided (an "S&P Deemed Rating" Confirmation"). For purposes of running the S&P CDO Monitor Test in connection with such Effective Date Report, (A) the Effective Spread will be calculated by assuming that any LIBOR Floor Obligation bears interest at a rate equal to (x) the stated interest rate spread of such Collateral Debt Obligation plus (y) the relevant London interbank offered rate-based index (without any adjustment to account for a London interbank offered rate floor) [Reserved; not applicable as of Amendment Effective Date and (B) the Monitor Principal Amount shall exclude the amount of proceeds on deposit in the Principal Collection Account and Unused Proceeds Account which are permitted to be designated as Collateral Interest Collections pursuant to Section 10.2(o) (clauses (A) and (B), together, the "S&P CDO Monitor Formula Adjustments"). If the Collateral Manager does not expect that an S&P Deemed Rating Confirmation will occur, including due to the S&P CDO Model Election Date having occurred as of or prior to the last day of the Ramp-Up Period, the Collateral Manager (on behalf of the Issuer) shall provide to S&P the Excel Default Model Input File and the LoanX ID of each Collateral Debt Obligation (if any), with a request that S&P reaffirm its Initial Rating of the Secured Notes.

the term "Cov-Lite Loan", (12) a First-Lien Last-Out Loan or (13) a Below-Par Collateral Debt Obligation; and

(D) an indication as to whether each such Collateral Debt Obligation utilizes a <u>LIBORSOFR</u> Floor (whether or not currently in effect), together with the value of any such <u>LIBORSOFR</u> Floor;

## (xx) [Reserved];

(xxi) During the S&P Rating Period, the identity of each Collateral Debt Obligation which has an S&P Rating derived from a Moody's Rating;

(xxii) The identity of each Current Pay Collateral Debt Obligation, the Market Value of each such Current Pay Collateral Debt Obligation, and the percentage of the Aggregate Collateral Balance comprised of Current Pay Collateral Debt Obligations;

(xxiii) The identity of each Deferrable Collateral Debt Obligation, the Market Value of each such Deferrable Collateral Debt Obligation, and the date on which interest was last paid in full in Cash thereon;

(xxiv) The details of any A/B Exchange and any Exchange Transaction consummated since the Report Determination Date of the prior Monthly Report or Note Valuation Report;

(xxv) The amount of any Contribution made since the previous Report Determination Date and on or prior to the current Report Determination Date (if any) and whether such Contribution (or portion thereof) is a Cure Contribution. For the avoidance of doubt, each Monthly Report does not reflect Contributions received after the related Monthly Report Determination Date in any manner;

(xxvi) The aggregate amount of all Cash deposited in the Permitted Use Account and the usage thereof since the Report Determination Date of the prior Monthly Report;

(xxvii) The (A) Total Diversity Score (during and after the Reinvestment Period), (B) Average Debt Rating (during and after the Reinvestment Period), and (C) Weighted Average Life (during and after the Reinvestment Period);

(xxviii) The identity of any purchase, sale or other transaction entered into pursuant to Article 12 involving an asset held by a fund or other investment vehicle (other than the Issuer) for which the Collateral Manager provides administrative and advisory functions;

(xxix) The details of any transaction entered into pursuant to Section 12.2(b)(i) hereof since the Report Determination Date of the prior Monthly Report

Affiliate of any Member, nor shall any Member be deemed to be an Affiliate of any other Member, solely by reason of such Member's control of the Company).

"Agreement" means this Limited Liability Company Agreement, as amended, modified or supplemented from time to time.

"Business Day" means any day other than a Saturday, Sunday or a day on which commercial banks are authorized or required to close in New York, New York.

"Certificate" means the Certificate of Formation of the Company, as amended, modified or supplemented from time to time.

"Company" means the limited liability company formed by the filing of the Certificate and governed by this Agreement under the name "[NAME], LLC."

"Control" means the possession, directly or indirectly, of the power to direct the management or policies of a Person, whether through ownership or voting of securities, by contract or otherwise.

"<u>Dryden Member</u>" means Dryden 87 CLO, Ltd. or any successor thereto in accordance with Article 8.4.

"Fiscal Year" has the meaning set forth in Section 6.3.

"Indemnified Party" has the meaning set forth in Section 7.8(a).

"Indenture" means the Indenture dated as of July 7, 2021 among Dryden 87 CLO, Ltd., as issuer, Dryden 87 CLO, LLC, as co-issuer, and U.S. Deutsche Bank National Association Trust Company Americas, as trustee, as amended from time to time.

"<u>Interest</u>" when used in reference to an interest in the Company, means the entire ownership interest of a Member in the Company at any particular time, including its interest in the capital, profits, losses and distributions of the Company.

"Material Action" means to institute proceedings to have the Company be adjudicated bankrupt or insolvent, or consent to the institution of bankruptcy or insolvency proceedings against the Company or file a petition seeking, or consent to, reorganization or relief with respect to the Company under any applicable federal or state law relating to bankruptcy, or consent to the appointment of a receiver, liquidator, assignee, trustee, sequestrator (or other similar official) of the Company or a substantial part of its property, or make any assignment for the benefit of creditors of the Company, or admit in writing the Company's inability to pay its debts generally as they become due, or consent to substantive consolidation of the Company with the Dryden Member or any Affiliate of the Dryden Member, or sell, exchange or transfer substantially all the assets of the Company, or take action in furtherance of any such action.

"Members" means the Dryden Member and the Special Independent Member (and "a Member", "any Member" and "either Member" means any of the Members).